See discussions, stats, and author profiles for this publication at: https://www.researchgate.net/publication/324523396

SINIS: A GQM+Strategies-based Approach for Identifying Goals, Strategies and Indicators for IT Services

Article in Information and Software Technology · April 2018

DOI: 10.1016/j.infsof.2018.04.006

TION		DEADE
ATION	5	reads 118
tho	×	
iiiio	13.	
	Bianca Trinkenreich	Gleison Santos
	Universidade Federal do Estado do Rio de Janeiro (UNIRIO)	Universidade Federal do Estado do Rio de Janeiro (UNIRIO)
	13 PUBLICATIONS 26 CITATIONS	133 PUBLICATIONS 465 CITATIONS
	SEE PROFILE	SEE PROFILE
	Monalessa Perini Barcellos	
\$	Universidade Federal do Espírito Santo	
	76 PUBLICATIONS 295 CITATIONS	
	SEE PROFILE	
	SEE PROFILE	

Some of the authors of this publication are also working on these related projects:

Project

Era da Transparência: A construção de um Modelo de Maturidade, Métodos e Ferramentas que possam apoiar as organizações na implantação de uma Arquitetura de Informações capaz de enfrentar este desafio View project

INTEROPERABILIDADE SEMÂNTICA DE INFORMAÇÕES EM SEGURANÇA PÚBLICA View project

Information and Software Technology xxx (xxxx) xxx-xxx



Contents lists available at ScienceDirect

Information and Software Technology



journal homepage: www.elsevier.com/locate/infsof

SINIS: A GQM + Strategies-based approach for identifying goals, strategies and indicators for IT services

Bianca Trinkenreich^a, Gleison Santos^a, Monalessa Perini Barcellos^b

^a Graduate Program in Computer Science, Department of Applied Informatics, Federal University of the State of Rio de Janeiro (UNIRIO), Rio de Janeiro, Brazil ^b Ontology and Conceptual Modeling Research Group (NEMO), Computer Science Department, Federal University of Espírito Santo (UFES), Vitória, Brazil

ABSTRACT

Context: Measurement is a key process to support organizations in managing and improving processes, products and services. The literature on IT (Information Technology) Service states that IT services should support critical business processes and should be measured in order to provide useful information for decision-making. However, there is a lack of clear guidance regarding what should be measured and which critical business processes should be considered.

Objective: We conducted this work aiming to answer the research question: How to support identification of IT service goals, strategies and indicators at different organizational levels and aligned to business goals?

Method: We defined SINIS, a method to identify Goals, <u>S</u>trategies and <u>In</u>dicators for <u>IT S</u>ervices, which has been developed to support IT service departments in identifying IT service goals, strategies and indicators to provide information for decision-making at different organizational levels and in alignment with business goals. SINIS supports defining strategies to achieve IT service goals, and identifying indicators to evaluate the strategies and goals achievement. SINIS is based on process improvement approaches (mainly GQM+Strategies) and approaches related to IT service management (mainly COBIT Goals Cascade)

Results: SINIS was used in a case study in the IT Infrastructure and IT Security departments of a large global company. Results showed that participants were able to build the GQM+Strategies Grid and discard useless indicators. In addition, team members became more devoted to measurement and strategies, and better understood relations between goals, strategies and indicators. Templates, examples and checklists useful to learn how to execute SINIS and to properly record the produced results were used by the study participants.

Conclusions: Initial evidences show that SINIS supports building the GQM+Strategies Grid and helps IT service departments to define strategies and identify useful indicators, contributing to focus efforts on strategies aligned to IT service and business goals.

1. Introduction

The service sector (involving information, health, education, tourism, entertainment, and others) has been recognized as the largest economic sector in developed countries and as an expanding sector in emerging markets [47]. This scenario has led organizations to shift from traditional production-based business models to new service-based ones [17]. In this sense, many organizations have started to use IT (Information Technology) to build service capabilities into their products, in order to be able to provide more accurate and faster service to customers [36].

Services involve delivering value to customers by facilitating the results they want to achieve without forcing them to take on the costs and risks of ownership. IT service management is a set of specialized organizational capabilities for providing value to customers through services. Its practice has been growing by adopting an IT management service-oriented approach to support applications, infrastructure and processes [39].

Guidance on how to implement and improve IT service practices is a key factor to improve service performance and customer satisfaction [19]. The use of suitable measures can help to monitor processes executed for delivering services and to support initiatives to improve managing IT service-related processes. Measurement activities can help organizations to monitor the performance of their projects and processes, leading to a high-maturity scenario. Thus, it is necessary to identify the processes to be measured and the measures to be used.

Effective service measurements should cover meaningful indicators,¹ in order to be able to verify goals achievement [11,16]. In

E-mail addresses: bianca.trinkenreich@uniriotec.br (B. Trinkenreich), gleison.santos@uniriotec.br (G. Santos), monalessa@inf.ufes.br (M.P. Barcellos).

https://doi.org/10.1016/j.infsof.2018.04.006 Received 6 June 2017; Received in revised form 3 April 2018; Accepted 13 April 2018 0950-5849/ © 2018 Elsevier B.V. All rights reserved.

¹ In this work, we use the indicator definition given by Barcellos et al. [8–11], who state that an *indicator* is a measure directly used to monitor goals achievement.

B. Trinkenreich et al.

this sense, the selection of the processes to be measured and the measures to be used should be aligned with organizational goals, so that measurement results can provide relevant information for decisionmaking and business support.

The IT service literature does not provide clear guidance or strict suggestions about which processes and measures should be considered for measurement. Moreover, properly identifying measures and indicators is not a trivial task. Even if a set of measures is available, it is still not easy to identify the proper measures and identify indicators for IT services [42]. Several factors contribute to the difficulty in defining measures and indicators for IT services, including: (i) lack of approaches to guide the definition of IT service indicators, (ii) lack of practical examples involving IT service indicators, and (iii) lack of measurement capabilities in IT supporting tools [27,33].

In the Software Engineering area, there are several proposals to aid organizations in measuring and improving software processes. GQM + Strategies [6] is a goal-oriented measurement approach that supports identifying goals at different organizational levels, defining strategies (i.e., initiatives, such as projects or other actions) to achieve the goals, and identifying measures/indicators to monitor strategies and goals. By establishing strategies aligned to organizational goals, it is possible to identify the processes involved, and measure and improve them. In addition, teams can focus on initiatives truly related to organizational goals, and understand how their work is aligned to those goals and how their work will be measured.

In view of the above, we developed *SINIS* (a method to identify Goals, <u>Strategies and Indicators for IT Services</u>), which applies principles from GQM+Strategies to help organizations define proper goals, strategies and indicators for IT services derived from and aligned to business goals. SINIS proposes a process which guides the steps to be followed by organizations in order to identify IT service goals, strategies and indicators. In addition to the process, SINIS provides a set of templates, checklists and examples to help organizations perform the SINIS process.

SINIS can help organizations which are just starting IT service measurement as well as organizations which have already started it and want to review or improve their goals, strategies or indicators. In the first scenario, SINIS can guide organizations on defining IT service goals, strategies and indicators aligned to their organizational goals. In the second scenario, SINIS can help organizations to review the defined IT service goals, strategies and indicators, and identify those aligned to organizational goals. SINIS can also help these organizations to define new IT service goals, strategies and indicators, if necessary.

To develop SINIS, we started by performing incremental learning cycles (five in total). Each cycle consisted of an investigative study performed to obtain knowledge relevant for developing SINIS. Five studies were performed. First, we carried out a systematic literature review to identify measures suitable for IT service measurement [49]. This study provided a set of measures which can be used as input to SINIS activities. Second, we performed a case study at a large company to investigate aspects which should be considered when identifying IT service indicators (for instance, the same indicator may be related to several IT services) [48]. Third, we performed a case study at an organization to evaluate the measures identified in the systematic literature review, and to identify new measures used by the organization. This study also contributed to SINIS development, because it allowed us to improve the set of measures identified in the systematic review. In this study, we also investigated impacts among IT service-related processes [50]. Understanding these impacts is important to establish proper strategies for achieving IT service goals. Fourth, we performed an action research involving the use of GQM+Strategies to identify indicators to IT service-related processes [51]. This study showed us that GQM+Strategies could be applied in the IT service domain, but some extensions would be necessary. Finally, we performed a qualitative study followed by a case study to investigate how to define or elicit strategies to achieve IT service goals [54]. The use of causal analysis

Information and Software Technology xxx (xxxx) xxx-xxx

techniques was found to be a way of identifying aspects on which strategies for achieving IT service goals should be focused.

SINIS was developed by following the Design Science Research method and involving two design cycles. In the first design cycle, we produced SINIS first version to be used by organizations starting IT service measurement (the first scenario previously mentioned). We conducted a case study to evaluate the use of this first version in the IT Infrastructure department of a large company [52]. This version considered only the scenario where organizations are interested in creating IT service goals, strategies and indicators from scratch. The case study results revealed limitations and opportunities for improvement, such as the need for checklists to support the execution of SINIS activities, and the need for changing SINIS activities to help organizations that have already started IT service measurement and want to review or improve goals, strategies or indicators (the second scenario previously mentioned). Based on these results, in the second design cycle we evolved SINIS to a newer version, which is presented in this paper. We made changes to SINIS activities and we also defined a set of checklists to support each SINIS activity. The improvements made allow organizations not only to start defining goals, strategies and indicators from scratch, but also to review goals, strategies and indicators in place, having a chance to improve them and the relations between them, include new ones or discard those not aligned to any goal. After developing the new version of SINIS, we conducted a case study applying it in the IT Security department of a large company. The new version of SINIS and the study performed to evaluate it are addressed in this paper.

The remaining of this paper is structured as follows: Section 2 provides the background for the paper, Section 3 presents the research method used to create SINIS, Section 4 introduces SINIS, Section 5 addresses the case study in which SINIS was used, Section 6 presents conclusions, related works and implications.

2. Background

There are several definitions of service. At a certain level, these definitions generally reflect the point of view of academic disciplines or economic sectors. According to OGCa [39], a *service* is "a logical representation of a repeatable activity that has a specified outcome. It is self-contained and is a 'black box' to its consumers." *IT services* (e.g., software application services and network services) support business services realization [13]. Thus, they are an important means towards establishing Business-IT alignment [1,23].

Service quality is an abstract concept due to the nature of the notion of service, which is intangible and non-homogeneous, and whose consumption and production are inseparable. It is a measure of how much a service level meets or does not meet customers' requirements and expectations. The intangibility of services makes it difficult to understand how customers observe and evaluate service quality [41].

To be able to offer quality, suppliers should continually assess how they are providing services as well as the future expectations of customers. Providing consistent quality is important, but is also one of the most difficult aspects of the service industry (ISO/IEC 20000, [26]). In order to assess and improve service quality, suppliers need to evaluate the quality of the processes they perform to deliver services. Guidance on how to develop and improve maturity practices for IT services is a key factor to improve service performance and customer satisfaction [19]. There are several models and standards addressing processes related to service, such as CMMI-SVC [19], ITIL [39], and ISO/IEC 20000 (ISO/IEC, [26]). All of them require the use of measures to monitor service-related processes.

Measurement plays a key role in process improvement initiatives [18,33] and can quantitatively demonstrate quality, predict processes behavior, and allow suppliers to increase the probability of achieving the expected IT service quality [37,45]. The basic element for measurement is measure, which quantifies aspects of entities to characterize

them. When data are collected for measures, it is possible to analyze them and obtain useful information. When information provided by a measure can be used to monitor goals achievement, measure is said to be an indicator [8–11].

In the literature, there are several works addressing aspects related to measurement and IT services. SINIS was developed based on the knowledge provided by some of them, namely: GQM+Strategies [6]; COBIT Goals Cascade [25]; Balanced Scorecard [29]; IT service standards such as ITIL [39], ISO/IEC 20000 (ISO/IEC, [26]), CMMI for Services [19]; and the Reference Software Measurement Ontology [11]. GQM+Strategies and COBIT Goals Cascade served as the main basis for SINIS. Therefore, these approaches are briefly presented in the following.

2.1. GQM + Strategies

The *GQM*+*Strategies* approach for goal-oriented measurement [6] is an extension of the Goal-Question-Metric approach [5], or GQM. It supports deriving, linking and disseminating goals and strategies across various levels of an organization, and helps verify the success or failure of strategies and goals using a measurement system. In GQM+Strategies, the *strategies* term refers to projects, actions or initiatives performed to achieve goals.

GQM + Strategies provides a model that relates goals and strategies at various organizational levels. One or more *strategies* can accomplish a *goal*. Context factors and assumptions influence goals and strategies. *Context factors* represent known organizational environment variables. *Assumptions* are predicted, estimated or guessed unknowns, which can impact the interpretation of measurement data, as well as associated goals and strategies [6].

GQM + Strategies supports not only the identification of goals and strategies at various levels of the organization, but also the definition of measures aligned to the organization's goals, and measurement data interpretation and compilation at each level [31]. For the latter purposes, GQM is used. A *GQM model* consists of a measurement goal plus associated questions, measures, and supplementary interpretation models. At each level, for each goal, a GQM model measures goals achievement considering the related strategy [6].

GQM + Strategies elements² and related GQM models are organized into a GQM + Strategies Grid, a graphical representation showing goals and strategies at each organizational level, including GQM models required for monitoring and verifying goals achievement. The grid makes goals and strategies explicit, as well as the related measurement initiatives. In addition, at each organizational level it is possible to get a clear understanding regarding how the strategies at that level contribute to goals at higher levels.

2.2. COBIT goals cascade

COBIT (Control Objectives for Information and related Technology) [25] is a good-practices framework created to support enterprise IT governance and management. *COBIT Goals Cascade* [25] is the mechanism to use stakeholder needs to derive enterprise goals, IT-related goals and enabler goals, which should be specific, actionable and customized. This derivation allows setting specific goals at each level of the organization, aligned to business goals and stakeholder requirements.

COBIT Goals Cascade is composed of four steps to cascade goals from top level (e.g., enterprise goals) to lower levels (e.g., IT-related goals), and then, similarly to strategies in GQM+Strategies, COBIT Goals Cascade introduces *enablers* as a way to achieve IT goals. The four steps of COBIT Goals Cascade are: stakeholder drivers influence stakeholder needs, stakeholder needs cascade to enterprise goals, enterprise goals cascade to IT-related goals, and IT-related goals cascade to enabler goals.

COBIT Goals Cascade provides a reusable catalog with 17 enterprise goals, 17 IT-related goals, 37 processes and more than 100 indicators [25]. Organizations can reuse these elements. For example, an organization with the enterprise goal "Business service continuity and availability" could use the COBIT Goals Cascade catalog and search for ITrelated goals related to this enterprise goal. It could select the IT-related goal "Adequate use of applications, information and technology solutions," and then search for processes related to this IT-related goal (e.g., "Manage Change Acceptance and Transitioning process") and indicators to measure the processes (e.g., "Number or percent of releases that fail to stabilize within an acceptable period"). However, as different market situations and environments require different measures, COBIT Goals Cascade recommends that each organization should build its own goals cascade, compare it with COBIT's, and then refine it [25].

COBIT is considered a framework to address IT governance challenges in a holistic way [3]. However, industry considers implementing COBIT Goals Cascade to be a difficult task, and it may be applied selectively [4]. In this sense, Steuperaert [46] proposed an improvement to COBIT Goals Cascade, which introduces an additional, limited set of Enterprise Strategies associated with IT-related processes.

3. Research method

The primary research approach adopted to develop SINIS was Design Science Research, which concerns on creating and evaluating new IT artifacts to help organizations to address important informationrelated tasks [22]. We used this research approach because the object of study is an artifact – specifically, a method including a process and instruments (checklists, templates and examples) to support the execution of that process – that aims at helping organizations to identify IT service-related goals, strategies and indicators. For considering the artifact context of use, its development should involve cycles of experimentation in the IT service industry.

According to HEVNER [22], Design Science Research is an iterative process including three cycles: the *Relevance Cycle*, the *Design Cycle* and the *Rigor Cycle*.

A Design Science Research project begins with the *Relevance Cycle*, which involves defining the problem to be addressed, the research requirements, and the criteria to evaluate research results. The problem addressed by the present research project involves the need of IT service organizations or departments to define suitable indicators to monitor IT service goals aligned to organizational goals. Considering the identified problem, we decided to develop a method to guide organizations in identifying IT service indicators by using organizational goals to derive IT service goals and strategies. We defined five requirements for SINIS:

(R1) Allow identifying indicators and strategies at different organizational levels to facilitate reporting the right information to each decision-making management level;

(R2) Foster alignment between indicators and business goals at those different levels;

(R3) Provide procedures and instruments (such as checklists, templates and examples) to support execution;

- (R4) Allow reusing indicators; and
- (R5) Adopt a consistent measurement terminology.

These requirements were established based on aspects indicated in the literature. Requirements R1 and R2 were defined based on [6], who emphasize that measurement should be aligned to organizational goals and cascaded to several organizational levels to provide useful information for each of them and for the organization as a whole. In addition, those authors advocate using strategies as a way of achieving the established goals, and implementing a measurement system to verify the achievement of goals and the success or failure of strategies.

 $^{^{2}}$ A GQM + Strategies *element* is composed of an organizational goal, the strategies related to it, and the context factors and assumptions that influence them [6].

B. Trinkenreich et al.

Information and Software Technology xxx (xxxx) xxx-xxx



Fig. 1. Overview of the design science research cycles in this work - based on [22].

Requirement R3 was based on studies which recommend the use of instruments to support execution of measurement-related activities. Basili et al. [6] suggest using questions for eliciting goals and categories for classifying goals, while Petersen et al. [40] suggest semi-structured interviews as a complement to templates, as well as notations for obtaining information regarding goals, strategies or indicators. Requirement R4 was based on [28,27,30], who state that identification of indicators can take a long time and suggest that reusing existing indicators. Finally, Requirement R5 considered works by Garcia et al. [20] and Barcellos et al. [8–11], which discuss problems in measurement terminologies and point out the need to use a consistent terminology to promote mutual understanding.

Established requirements should be considered as a criterion to evaluate the proposed method. Moreover, it should be evaluated to what extent SINIS can be used to support identifying goals, strategies and indicators for IT services. For that, *feasibility* and *usefulness* of the method should be considered. SINIS should be considered *feasible* if it can be executed according to its description, if it produces what it is supposed to deliver (IT service-related goals, strategies and indicators) and if its execution requires efforts considered acceptable. On the other hand, SINIS should be considered *useful* if SINIS application provides benefits for the organization. Thus, we defined feasibility and usefulness indicators as follows: *feasibility* = applicability & efficacy & acceptable effort (applicability = SINIS can be executed according to its description; efficacy = SINIS produces what it is supposed to deliver; acceptable effort = SINIS execution requires efforts considered acceptable) and *usefulness* = SINIS use provides benefits for the organization.

The *Design Cycle* involves developing and evaluating artifacts or theories to solve the identified problem. In the present work, the proposed artifact is SINIS, which was developed through two design cycles, each one of them resulting in a version of SINIS. The first design cycle resulted in the version presented in [52]. The second design cycle resulted in the current version, which is presented in this paper. SINIS evaluation was carried out through two case studies. First version of SINIS was evaluated through a case study in an IT Infrastructure department of a large company and provided results for us to improve SINIS [52]. The current version of SINIS was evaluated through another case study in an IT Security department of the same company. In order to develop SINIS, as discussed in the Introduction, we performed five incremental learning cycles, consisting of investigative activities to gain

useful knowledge to create SINIS.

Finally, the Rigor Cycle refers to using and generating knowledge. Rigor is achieved by appropriately using foundations and methodologies from a knowledge base grounding the research, and adding knowledge generated by the research to contribute to the growing knowledge base. In this work, the main foundations are knowledge related to IT services, measurement, systematic literature review and evaluation methods such as case study. Several pieces of the knowledge used to develop SINIS were obtained through the five incremental learning cycles performed. Certain sources were particularly useful for developing SINIS, with pieces of knowledge which they provided being incorporated in SINIS: principles from GQM + Strategies [6] and Process Analysis [15] were used to support defining indicators and strategies aligned to business goals at different organizational levels, contributing to satisfy R2 and R3; ideas from COBIT Goals Cascade [25], Balanced Scorecard [29], ITIL (OGCa, [39]), ISO/IEC 20000 [26], and CMMI for Services [19] were used to develop checklists, templates and examples provided by SINIS to support its execution, contributing to satisfy R3 and R4; and the Reference Software Measurement Ontology addressed in [8-11] provided the measurement conceptualization and terminology adopted in SINIS, contributing to satisfy requirement R5.

The main contribution to the knowledge base is SINIS itself, as a new method to support the identification of goals, strategies and indicators for IT services. Additionally, the investigations performed in the five incremental learning cycles also contributed to the knowledge base, providing: (i) a set of measures suitable for IT service measurement [49]; (ii) aspects to be considered when identifying IT service indicators [48]; (ii) knowledge related to analysis of impacts among IT service-related processes [50]; (iii) relevant knowledge about using GQM+Strategies in the IT service domain [51]; and (iv) knowledge involving the use of causal analysis techniques to identify aspects to be addressed when defining strategies to achieve IT service goals [54]. Lastly, the studies carried out to evaluate SINIS in industry can serve as examples for other people to apply SINIS.

Fig. 1 summarizes the main information related to the Design Science cycles in this research. As shown in the figure, there are intersections between the Relevance and Rigor cycles and the Design Cycle – i.e., the Design Cycle takes into account the Relevance Cycle (e.g., the method should meet the established requirements) and the Rigor Cycle (e.g., the development of the method should be grounded in scientific theories and methods).

3.1. Threats to validity

There are some threats involved in the way we developed SINIS. First, the use of various existing approaches, libraries, frameworks and techniques from the literature as a basis to build SINIS could result in a method hard to be understood and used. In order to minimize this threat, we defined SINIS as a systematic process to be used as a step-bystep procedure, and we created a set of supporting instruments (checklists, templates and examples) to help the users on performing SINIS activities. By doing that, previous knowledge of approaches, libraries, frameworks and techniques used as a basis to SINIS is not required. Moreover, the use of these several bases to SINIS could result in a method that fails to achieve its requirements and it would be bad to found out that only after the method is created. To treat this threat, we adopted a development approach based incremental learning cycles. This way, each of the basis to SINIS was evaluated in the IT Service domain before being incorporated to the method.

In the context of the incremental learning cycles there is a threat that comes from the fact that some studies performed in the incremental learning cycles were performed in the same company. This could result in a method designed for a single-case scenario. In order to minimize this threat, we performed the studies in different departments, involving different people. Besides, we created supporting instruments based on specialized literature instead of on particular characteristics of the organization. By doing that, we provide a more general solution that can be used by other organizations. Moreover, we considered in the method scenarios involving organizations that do not have any defined goals, strategies or indicators, as well as organizations that have already started measuring, but need to improve their goals, strategies or indicators. This amplifies the range of organizations that can benefit from using SINIS.

Concerning construct validity, which is related to the measures used to evaluate the object of study, the main threat refers to the indicators used to evaluate SINIS, since they may not be able to fully represent the properties they operationalize. To minimize this threat, we defined indicators for the properties we considered necessary to SINIS be applied to solve the problems it was conceived to solve. These properties were used as acceptance criteria in SINIS evaluations.

As for internal validity, the main threat is that all studies performed during SINIS development were conducted by researchers, including the two case studies performed to evaluate SINIS. The researchers' participation can have affected the studies' results (different results could be obtained if the researchers have not intervened in the studies). Trying to minimize this treat, researchers' participation was limited to activities in which their intervention was really necessary (for example, researchers performed themselves all activities in the systematic mapping, while in studies involving IT service departments, researchers limited to observe or guide activities' execution). Additionally, except by the last case study (discussed in Section 5), data obtained from studies were analyzed and interpreted by more than one researcher.

The strategy we adopted to evaluate SINIS is case-based. One of the biggest threats in this context is the ability to generalize from the casespecific findings to different cases [56]. Thus, the main threat to external validity in this study is about results' generalization. Generalizations produced in Design Science Research are considered middle range, generalizing beyond the case level, but not intending to be universal [55]. They are valid to restrict ranges of contexts and, although are not universal, they are more useful in practice [56]. In casebased research, after getting results from specific case studies, generalization can be established for similar cases. For evaluating SINIS, we selected a large organization with IT service culture and well-established IT-related service departments and processes. We considered that scenario favorable to apply SINIS and a good one to provide results that could be also expected in other similar organizations. We performed two case studies in different IT service-related departments of this same organization. The first version of SINIS was applied in the IT

Information and Software Technology xxx (xxxx) xxx-xxx

Infrastructure department [52] and the second version, the one addressed in this paper, was applied in the IT Security department. Thus, SINIS results cannot be exhaustively generalized. However, we believe that it is plausible that they can be generalized for similar environments.

4. SINIS – a method to support identifying goals, strategies and indicators for IT services

SINIS is a method to support organizations in identifying IT servicerelated goals, strategies and indicators in alignment with organizational goals. The method allows identifying strategies that contribute to goals achievement and indicators that can be used to monitor strategies and goals. In this work, the strategy term is used as proposed in GQM + Strategies [6]. Thus, it refers to projects, initiatives or actions defined or performed aiming at goals achievement.

As discussed in the previous sections, SINIS reuses knowledge provided by other proposals, being based mainly on GQM + Strategies [6] and COBIT Goals Cascade [25]. In fact, SINIS could be understood as an extension of GQM + Strategies for the IT service domain. SINIS extends GQM + Strategies by addressing aspects that are particular to IT services, such as IT service goals and indicators. Moreover, different from GQM + Strategies, SINIS provides guidance on how to identify processes to be considered in strategies and also defines a step-by-step procedure supported by checklists, templates and examples that help organizations to use the method. In SINIS, like COBIT Goals Cascade and GQM + Strategies, goals are cascaded from organizational top levels (organizational goals) to lower levels (IT service goals and strategies).

SINIS consists of a process comprising a set of activities to identify (or define) relevant IT service goals, strategies and indicators, along with a set of templates, checklists and examples to support performing these activities. Considering that SINIS involves IT service and measurement-related activities, we suggest that it should be applied by personnel familiar with IT service processes and measurement; however, no knowledge of the approaches used as foundations for SINIS (e.g., COBIT Goals Cascade and Balanced Scorecard) is required.

SINIS can help organizations which are just starting IT service measurement as well as organizations which have already started it and want to review or improve their goals, strategies or indicators. Both scenarios are covered by activities listed below and we indicate to skip any activity when applicable. Fig. 2 shows an overview of SINIS. The SINIS process is composed of five phases. In the figure, the activities of each phase are identified with the same color. Below we describe SINIS phases and activities, and also present some of the templates which may be used when performing the activities. A full description of the SINIS activities plus all the checklists, templates and examples provided to help SINIS use is available at [53].

SINIS goal is to help IT service organizations to:

- Define IT service goals, aligned with business goals;
- Define strategies for people (or teams) to work on aiming at IT service goals achievement;
- Define indicators and measurement plans to measure the achievement of IT service goals and strategies;
- Define interpretation models for indicators.

4.1. Elicit context factors and assumptions for IT services

This phase involves obtaining relevant information about the organization and the IT service department at which SINIS will be used that can influence the selection and definition of goals, strategies and indicators.

IT service goals and strategies are defined within the context of the organization, where options are limited by the capabilities, issues or constraints of the organization. Thus, it is necessary to identify context factors and assumptions. Context factors are aspects which are factually

Information and Software Technology xxx (xxxx) xxx-xxx



Fig. 2. Overview of SINIS.

known (e.g., the organization needs to improve service availability) and assumptions are aspects which are believed to be true, but with little or no evidence (e.g., in the organization, IT service costs cannot be increased). Context factors and assumptions provide useful information to define the scope of the IT service goals and strategies. In addition, differentiating between what we factually know and what we believe to be true will help later to properly interpret measurement data and find potential reasons for strategies which did not succeed [6].

Documents (such as service level agreements and contracts) can be used as a source for identifying context factors and assumptions. If documents are not available, meetings with stakeholders can be used to gather information. Examples of questions which can be used to elicit context factors and assumptions are presented in Table 1. Context factors and assumptions can be recorded by using the template shown in Table 2.

4.2. Define IT service goals and indicators

This phase involves gathering, defining and reviewing IT service goals based on organizational goals. In addition, it addresses defining indicators to monitor the identified IT service goals. It is composed of four activities, which are described below.

(i) Select IT Service Goals: During this activity, existing IT service goals are reviewed to select those aligned to organizational goals. Information about IT service goals can be obtained from analyzing documents or interviewing IT service department managers. If the organization has not defined IT service goals, it must skip this activity and

Table 2 SINIS template for context factors and assumptions.

Context Factors		
Context factor	< Description of context factor – what is factually known – related to IT services >	
Context factor source	${<} {\rm Description}$ of the source from where the context factor was obtained ${>}$	
Date	< Date when context factor was considered >	
Responsible	< Person responsible for describing this context factor >	
Assumptions		
Assumption	< Description of assumption – what we believe is true but	
	have little or no evidence about – related to IT services $>$	
Assumption source	< Description of the source from where this assumption was obtained >	
Date	< Date when assumption was considered >	
Responsible	< Person responsible for describing this assumption $>$	

perform the next one.

SINIS provides a checklist with actions to be performed when identifying IT service goals, including: ensuring that the scope of a goal is related to something that the IT service department is able to deliver; verifying whether the service goal is actually related to an organizational goal that identifies which IT service process is related to the goal; classifying the goal as Maintaining, Increasing or Reducing something; quantifying the degree to which the goal should be achieved; defining a timeframe to achieve the goal; identifying the primary role responsible for the goal; and identifying the relevant constraints which may prevent

Table 1

	SINIS questions to support elicitation of IT Services Context Factors and Assumptions
1	What is the scope of the IT service department?
2	Which are the clients of the IT service department?
3	What information can be gathered from existing service level agreements and contracts with clients about the provided services?
4	What can be extracted from issues root cause on history reports about the provided services?
5	What can be extracted from customers satisfaction reports about the provided services?
6	What can be linked to IT services from organization vision, mission and values statements?
7	Which are the organization's goals?
8	What links can be identified between the IT service department and economic, social, political, market trends, and environmental and technological aspects
9	Which are the perceived IT services critical success factors?
10	What can be used from existent measurement models, data, baselines and targets?

Table 3

Template for recording IT service goals.

IT service Goal	
IT service Goal	<name goal="" it="" of="" service="" the=""></name>
Activity	< Is the goal to Maintain, Increase or Reduce? >
Object	< What is the object the goal is related to?>
Magnitude	< What is the quantity of the goal which should be achieved?>
Time Frame	< When should the goal be achieved?>
Responsible	< Who are the role primarily responsible for achieving the goal?>
Constraints < What relevant constraints may prevent achieving the goal? >	
COBIT Goals Cascade	
IT-related Goals	 < One of 17 available IT-related goals from COBIT:
	Aligning IT strategy and business strategy
	• Ensuring IT compliance and supporting business compliance with external laws and regulations
	• Ensuring commitment of executive management to IT-related decision-making

Managing IT-related business risk

- Realizing benefits from IT-enabled investments and services portfolio
- Ensuring transparency of IT costs benefits and risks

· Delivering IT services in alignment with business requirements

- · Adequately using applications, information and technology solutions
- Ensuring IT agility
- Ensuring security of information, processing infrastructure and applications
- · Optimizing IT assets, resources and capabilities
- · Enabling and supporting business processes by integrating applications and technology into business processes
- · Delivering programs providing benefits, on-time, on-budget, and meeting requirements and quality standards
- Providing reliable and useful information for decision-making

· Ensuring IT compliance with internal policies

- · Ensuring competent and motivated business and IT personnel
- Developing knowledge, expertise and initiatives for business innovation >
- <Finance, Customer, Internal, or Learning and Growth>
 - < Process which can impact goal achievement>

achieving the goal.

BSC Dimension IT service-related Processes

Table 3 shows the template for documenting an IT service goal. The template is based on GQM+Strategies [6] and requires information regarding the Balanced Scorecard (BSC) dimensions [29] related to the IT service goal. BSC dimensions are helpful in subsequent activities which involve reusing COBIT Goals Cascade indicators [25], since in COBIT Goals Cascade goals are classified by BSC dimension. In addition, it is relevant to identify IT service-related processes because measures associated with them might be useful as indicators [21].

(ii) Include New IT Service Goals: This activity consists in defining new IT service goals, following SMART principles: Specific, Measurable, Achievable, Relevant and Time-Bound [14]. In addition, goals should not be broad or vague. They should be broken down into specific results, written using words which clearly describe the results to be achieved as evidenced by indicators [12]. Examples of questions which can support defining IT service goals include the ones presented in Table 1 and also the following [6,40]: What does the IT service department plan to execute in the next period? What future is envisioned for the IT service department in the following years? How does the IT service department want to grow, gain new clients or provide services based on new competencies? What is the definition of success for the IT service department? Does the IT service department need or desire to improve any aspect of its service delivery?

(iii) Associate Existing Indicators to IT Service Goals: During this activity, the indicators in use by the IT service department are gathered and analyzed, and then it is verified whether indicators are associated with the IT service goals. Information about the indicators can be gathered from existing measurement documentation, repositories or from meetings. Indicators not related to any IT service goal should be discarded. For each indicator, a measurement plan should be established in order to allow properly understanding the indicator and make explicit its related goals. SINIS provides a template for documenting measurement plan for indicators, including the following information: related IT service goal related, measurement goal, information need, indicator, measurable entity type, base measures, measure calculation formula, measurement procedure, measurement responsible, measure unit, measurement moment and measurement periodicity.

(iv) Create New Indicators for IT Service Goals: During this activity, new indicators should be defined and associated to IT service goals. New indicators should be defined when the existing ones are not able to provide the necessary information, or when there is no indicator. Reusing indicators can help to reduce the effort, time and cost spent on this activity. In this sense, measures suggested in the COBIT Goals Cascade process [25] and in the IT service list of measures provided in [49] can be reused or can inspire defining new ones. The defined indicators should be described in measurement plans including the information cited in the previous activity.

4.3. Elicit strategies to achieve IT service goals, indicators and interpretation models

This phase involves identifying processes critical to IT services and root causes for the main issues to be addressed by strategies for achieving IT service goals. In addition, this phase addresses defining indicators and the respective interpretation models to monitor the defined strategies. It comprises five activities, described below.

(i) Gather Existing Strategies: During this activity, if the organization already has a list of IT service strategies planned or on course to achieve IT service goals, these strategies are gathered and reviewed. Information about the strategies can be gathered from documents, meetings records or by interviewing IT service managers and teams.

Strategies should be analyzed considering if their results contribute to achieve IT service goals. When the name or description of a strategy is not clear, it should be rewritten to clarify how it aligns with the related goals. However, when the results of an existing strategy do not contribute to goals achievement, the strategy should be reformulated or stop being executed, to avoid wasting efforts working on a strategy not aligned to any goal. Table 4 shows the SINIS template for documenting strategies.

(ii) Analyze Critical IT Service Processes: This activity consists of analyzing critical processes as a way to support defining strategies to achieve IT service goals. A critical process is a process which can impact business goals, a failed process, or a process which might fail [24]. Strategies should focus on critical processes or aspects that impact goals

B. Trinkenreich et al.

Table 4

SINIS template for strategies [52].

IT service Goals	<associated goals="" it="" service=""></associated>
Strategy Name	< Name of the strategy >
Strategy Scope	< Description of scope for the strategy >
Strategy Owner	< Personnel responsible for the strategy >
Strategy Sponsor	< Sponsor for funding the strategy >
Strategy Complexity	< Low, Medium, High>
Strategy Risk	< Low, Medium, High>
Strategy Cost	<cost perform="" strategy="" the="" to=""></cost>
Strategy Context Factors	< Context factors considered for this strategy >

achievement. Thus, SINIS recommends that critical processes related to IT service goals are identified and modeled to provide a detailed view about how the processes relate to goals and to provide insights for establishing strategies. In addition, relationships between processes should be investigated to identify critical cause-and-effect relationships which should be considered when establishing the strategies [51].

SINIS does not specify any technique or detail level to model processes. The important is that process modeling be sufficiently understandable to allow identifying possible conflicts, issues, difficulties or problems which might impact goals achievement and, thus, should be addressed by the strategies. Questions which can help identify critical processes include: Which processes related to IT service goals have the highest potential for improvement? Is there a plan for improving them? Which processes are currently not performing well? Which processes have strong correlation with goal results? Which processes are important for stabilizing performance? Which processes pose significant risks for a goal, or might prevent it from being achieved? Which processes serve as key inputs to a goal? Which processes consume much effort to be performed?

After gaining clear knowledge about the target processes, an investigation should be carried out to identify which part of the processes (i.e., sub-process) should be addressed by strategies. A root-cause investigation can help identify causes of problems in the critical processes and understand which part of the processes should be focused by the strategies. For performing causal analysis, several methods can be used such as document analysis, interviews, brainstorming, flowcharts, Five Whys, and Pareto, among others [43].

(iii) Establish Strategies to Achieve IT Service Goals: This activity involves determining what needs to be done to achieve IT service goals. During this activity, existing strategies are reviewed, and new strategies are defined in a round of brainstorming meetings conducted with the stakeholders. The purpose of these meetings is to relate strategies to IT service goals, discard useless strategies, adjust deficient strategies and define new ones. A same strategy can be related to different goals and more than one strategy can be necessary to achieve a goal.

SINIS advocates that strategies to achieve IT service goals should focus on processes which impact goals achievement (i.e., critical processes) and on solving difficulties which prevent goals achievement. Based on the results of the root-cause investigations performed in the previous activity, strategies to achieve IT service goals are defined. When defining the strategies, specific context factors and assumptions Information and Software Technology xxx (xxxx) xxx-xxx

should be elicited for each strategy. They must be consistent with the general context factors and assumptions previously defined in the Elicit IT service Context Factors and Assumptions phase.

(iv) Identify Indicators for Strategies: In this activity, similar to what is done to IT service goals, strategies are made measurable by specifying indicators to measure them. Indicators identified during the Associate Existing Indicators to IT Service Goals and Associate Existing Indicators to IT Service Goals and Associate Existing Indicators to IT Service Goals activities can also be associated to strategies defined to the goals the indicator relate to. Moreover, new indicators can be defined. For each indicator, a measurement plan should be established. In order to reduce effort, time and cost, SINIS recommends reusing existing measures by consulting measures suggested in the COBIT Goals Cascade [25] and in the list of IT service-related measures [49].

4.4. Create interpretation models for all indicators

In this phase, interpretation models for all indicators (related to IT service goals and strategies) are created. This activity is executed only at this moment to avoid rework and wasted time in case of discarding or adjusting any indicator. Interpretation models determine how collected data should be interpreted to support informed decisions about IT service strategies and goals achievement. Targets can be defined based on previous service level agreements and contracts, reports or business needs. Meetings with stakeholders can be used as a way to gather information. Questions to support obtain relevant information for defining interpretation models include: What is the expected result (or range) for each indicator in order to achieve the associated goal? If the result is above or below the range, should it be interpreted as good or bad? What was the last measured result for each indicator? Who are the personnel responsible for interpreting the indicator? How/when should the indicator be interpreted? Table 5 provides a template for recording interpretation models.

4.5. Build, review and adjust GQM + Strategies grid

During this phase, context factors, assumptions, goals, strategies and indicators are organized into a GQM+Strategies Grid in order to provide an overview of the results produced during the previous activities. The grid helps validate information and identify items which need to be reviewed. Flexibility is important to allow iterative changes in the GQM+Strategies Grid, to ensure that the overall plan is aligned with and reflects the organization needs [38]. Ideally, the grid should present the cleanest possible view of the defined IT goals, strategies and indicators. Fig. 3 shows the template that can be used to create the grid. SINIS suggests designing the grid in a way that goals, strategies and indicators at different levels can be viewed on a single page. In addition, general context factors and assumptions should also be included, making it possible to verify whether they are up-to-date. Thus, if there is a need to update the context factors and assumptions, the grid provides a convenient view of the goals, strategies and indicators which would be impacted by the changes.

The GQM+Strategies Grid interpretation models should be presented to all stakeholders in order to be validated and have the applicability, completeness, accuracy and consistency of the goals,

Table 5

Related Indicator	<name indicator="" of="" the=""></name>
Target	< Value desired for the indicator in order to achieve the associated goal >
Range	< Upper and lower values to be achieved by the indicator, defined based on historical data or organization goal $>$
Baseline	< Reference value that reflects a previously measured behavior of the indicator considering historical data >
Interpretation Model	< Procedure to be followed to analyze data collected for the indicator >
Interpretation Responsible	< Role performed by personnel in charge of analyzing the data >
Interpretation Moment Activity.during.which.the data analysis should be performed>	
Interpretation Periodicity	< Frequency at which the data analysis should be performed >



Fig. 3. SINIS Template for GQM + Strategies Grid.

strategies and indicators evaluated. In addition, discussions can point out potential findings and opportunities for improvement. It is recommended to involve personnel who were not involved in applying SINIS, but who will be eventually involved in or impacted by the execution or results of the defined strategies. During this phase, if any adjustment is needed, it is possible to go back to the activity in which the adjustments need to be made, and then continue applying SINIS again from that point to the end. For example, if an IT service goal needs to be adjusted, the related indicator, interpretation model, and strategies also need to be revisited.

5. Applying SINIS

As we discussed in Section 3, the strategy we adopted to evaluate SINIS is case-based. By following this strategy, we study individual cases and try to hypothesize a generalization for a similar population [56]. Thus, to evaluate the use of SINIS in a practical context, SINIS was applied in a case study to define IT service goals, strategies and indicators in an IT service department of a large company. The following sections present the case study design (Section 5.1), execution and data collection (Section 5.2), analysis, interpretation and lessons learned (Section 5.3), and threats to validity (Section 5.4).

5.1. Case study design

The *goal* of the case study was to evaluate the use of SINIS in a real context in industry as a way to support identifying IT service-related goals, strategies and indicators.

Considering this goal, the following *research question* should be answered:

(RQ1) To what extent can SINIS be used to identify goals, strategies and indicators for IT services?

The organization selected for the study is a large global organization headquartered in Brazil (referred here as "Organization A" to maintain confidentiality). This organization operates in over 30 countries and has offices, operations, exploration and joint ventures across five continents. SINIS was applied in the IT Security department. The IT Security department is part of the IT Services area that, in turn, is part of the Information Technology (IT) organizational unit of the Organization A. The IT Security department is headed by an IT Security Manager with the help of an IT Service coordinator, who coordinates the 15 members that compose IT Security team and acts as an interface between the manager and the team. IT Security team performs processes such as access management, vulnerability management, threat management, proactive monitoring and risk management. Some of the responsibilities of the IT Security department are: define the IT Security strategy, translating it into a roadmap of projects and initiatives towards the security and protection maturity defined by the leadership of the company; establish and monitor governance making sure the decisions on information security are being taken at the right level of the company; ensure the performance of IT Security processes and their related technological components by ensuring compliance with the policies, standards and processes established in the company; manage the agreed levels of services in the provision of IT security services; act on incidents and issues related to IT security; guarantee that employees are aware of the value of information becoming committed to their role in protecting it, reducing the likelihood of successful cyberattacks; and act as the center of excellence in cyber-security for the whole digital surface of Organization A, offering a catalogue of services for prevention and detection security operations.

The IT Security manager reported that his team was spending too much effort to perform measurements not aligned with the organization and the department goals. Thus, neither the coordinator nor team members did not know why they were spending time on measurement activities, leading them to lose motivation and confidence in the measurement results. Although many measures had been collected, the team members did not know how these measures were related to IT Security goals, or whom they should be reported to or how to interpret the measurement results. Moreover, the IT Security manager was asked by the board of directors of the organization to build a single-page report with a dashboard showing IT Security indicators. He was unsure of which indicators were more relevant to be included in the dashboard and would have to review the indicators and build a high-level dashboard. The study *participants* were the IT Security coordinator, the IT Security manager and the IT Security team.

The *procedure* planned for performing the study consisted of following the SINIS phases and activities, and using the provided templates, checklists and examples to define IT service goals, strategies and indicators to the IT Security department.

For answering the study research question, it is necessary to evaluate if SINIS is feasible and useful. As discussed in Section 3, SINIS should be considered feasible if it can be executed according to its description, if it produces what it is supposed to deliver (IT servicerelated goals, strategies and indicators) and if execution requires efforts considered acceptable. On the other hand, SINIS should be considered useful if its application provides benefits for the organization. Thus, after SINIS use, data collection should be done through interviews with participants to get feedback about SINIS. The research question should be used as the main question for the interviews and some supporting questions related to the research question should be used in the case the interviewee does not provide enough information based only on the research question. Examples of these supporting questions are: Did SINIS help to properly identify IT service goals? Did SINIS help to properly identify IT service strategies? Did SINIS help to properly identify IT service indicators? Did SINIS help to properly identify the relations between IT service goals, strategies and indicators? Did the produced grid help to understand the relations between IT service goals, strategies and indicators? Did the checklists, templates and examples help to perform SINIS activities? What benefits were perceived

from SINIS use? What problems or limitations were perceived when using SINIS? Do you consider the effort demanded to perform SINIS acceptable? Interviews should be recorded and transcribed in a document. *Data validation* should be done by showing the document containing the recorded data to the participants (personally or by email) and asking for their validation. After collecting and validating data, they should be interpreted (*data analysis*) aiming to verify if SINIS was considered feasible and useful. Thus, the answers provided by the interviewee(s) should be analyzed aiming to get information that allows verifying if it was possible to execute SINIS activities based on their descriptions, if SINIS produced what it is supposed to deliver, if its execution required efforts considered acceptable and if its use provided benefits for the organization.

5.2. Case study execution and data collection

In this section, we describe how SINIS activities were performed and present some results produced during SINIS application. The activities were conducted by the first author and performed together with the study participants. As suggested in SINIS, during some activities we performed meetings and semi-structured interviews, taking a maximum of 2 hours each. Some meetings/interviews involved the IT Security coordinator, while others included all team members (including the IT Security manager), and others the domain expert (a team member). After applying SINIS, we conducted an interview with the IT Security coordinator to obtain feedback about SINIS use.

5.2.1. Elicit IT service context factors and assumptions

In this phase, relevant context factors and assumptions were identified based on organizational goals and other information about the organization. Considering that the main organizational goal was to reduce costs, the first context factor identified was Organization main goal is to reduce cost and the related assumption was the IT Security department should not increase costs. Based on information provided by the IT Security manager, the second context factor identified was IT Security department does not have people dedicated to measurement activities and the related assumption was a member of the IT Security team is responsible for collecting and storing data for indicators and presenting the results to the manager at the weekly meeting.

5.2.2. Define IT service goals and indicators

As described in Section 4, this phase starts with the Select IT Service Goals activity. According to the IT Security manager, there was not any established IT service goal, since the team collected and reported indicators, not thinking about goals. Therefore, it was not possible to select IT service goals. Thus, the Define New IT Service Goals activity was performed. By using questions suggested by SINIS and conducting some meetings, the following IT service goals were identified:

Reduce the costs of resolving IT Security incidents;

- Reduce the time for resolving IT Security incidents;
- Increase the efficiency in executing controls;
- Reduce the number of users with elevated access to the Internet;
- Increase the efficiency of blocking malware messages;
- Reduce the number of users having SAP Segregation of Duties (SOD) conflicts;
- Increase the productivity of the IT Security team;
- Maintain the compliance of applications with IT Security policies;
- Increase the detection and remediation of vulnerabilities;
- Increase the efficiency of workstations and the protection of servers.

Each IT service goal was documented by using the template suggested by SINIS. An example is presented in Table 6.

Once the IT service goals were identified, it was necessary to Associate Existing Indicators to IT Service Goals. The IT Security

Table 6
IT service goal documented according to the SINIS template.

IT service Goal	Reduce the costs of resolving IT Security incidents	
Activity	Reduce	
Object	Costs associated with resolving IT Security	
	incidents	
Magnitude	10%	
Time Frame	Annual	
Responsible	IT Security Department	
Constraints	Do not increase cost	
COBIT Goals Cascade IT-related	Delivery of IT services in line with business	
Goal	requirements	
BSC Dimension	Customer	
IT service-related Process	Incident Management	

coordinator provided a spreadsheet with 39 indicators. For each indicator, the spreadsheet informed only its name and the role responsible for data collection, being, most of times, difficult to understand what the indicator really measured. Thus, it was necessary investigate the indicators meaning by asking the IT Security coordinator and team. Based on the obtained information, we changed the name of some indicators to make it clearer (e.g., the Profiles indicator was renamed to Number of Profiles Maintained by the Identity Access Management (IAM) System). Although the IT Security coordinator referred to the spreadsheet content as a list of indicators, it should be understood as a list of measures, because the measures were not aligned to any goal (the IT Security department did not have any established goal at that time). The relations between the indicators and the defined IT service goals were identified. For example, the indicator Percentage of incidents where field intervention was necessary to solve the issue (manual/ total) was associated to the IT service goals Reduce the cost of resolving IT Security incidents and Reduce the resolution time for IT Security incidents because resolve an incident manually tends to demand more effort and time and have a higher cost than resolve it automatically. Thus, a high number of incidents that had to be manually solved impacts the achievement of those goals. Seven indicators were discarded, because they did not relate to any IT service goal. Table 7 shows the indicators and the related IT service goals.

By using the template suggested by SINIS, a measurement plan was established for each indicator, containing information about the indicator and its relationship with IT service goals. An example is presented in Table 8.

After relating indicators to IT service goals and establishing measurement plans, the IT Security team decided that the selected indicators were enough. Thus, it was not necessary to perform the **Create New Indicators for IT Service Goals** activity.

5.2.3. Elicit strategies to achieve IT service goals, indicators and interpretation models

During this phase, strategies to achieve the IT service goals were established, as well as indicators to monitor the strategies. The first activity of this phase is Gather Existing Strategies. The IT Security coordinator reported that the department did not have documented strategies. In fact, this was expected, since at the beginning of this study the team informed that the goals they should achieve were not clearly defined. Considering that it was not possible to selected existing strategies, it would be necessary to define them. Aiming to identify the processes and aspects to be addressed by the strategies to be defined, we performed the Analyze Critical IT Service Processes activity. In this activity, we analyzed the processes related to the IT service goals and for some of them (according to the priority established by the IT Security coordinator and team) we developed the process model and investigated causes that could influence the IT service goals achievement. It is important to point out that the process modeling suggested by SINIS is not intended to be exhaustive. Its purpose is only to support discovering parts of the processes and issues that the strategies should

Information and Software Technology xxx (xxxx) xxx-xxx

Table 7

Selected indicators and related IT service goals.

Indicator	IT Service Goal
Number of opened audit findings	Increase efficiency in the execution of controls
Number of opened audit findings expiring next month	
Number of reported phishing cases	Increase efficiency in blocking of malware messages
Number of messages blocked by the gateway	
Number of virus events	Increase efficiency of workstations and protection of servers
Percentage of machines with antivirus up-to-date	
Percentage of machines without antivirus	
Antivirus (SLA) server	
Antivirus (SLA) workstations	
Percent of total number of threats that were solved/deleted/eliminated within one month	
Number of threats against servers or workstations that presented issues and could not be deleted or quarantined	
Number of profiles maintained by Identity Access Management system	Increase the productivity of the IT Security team
Number of application owners that participate in IAM (Identity Access Management) processes	
Number of legal or internal audit investigations supported by IT Security team	
Number of IT users	
Number of applications that use Identity Access Management system for authentication	
Number of found vulnerabilities unique on Organization A environment	Increase detection and remediation of vulnerabilities
Vulnerability management resolution rate	
Number of critical applications with critical vulnerabilities	
Number of servers and workstations scanned	
Number of IT Security exceptions accepted	Maintain adherence of applications to IT Security policies
Number of technical notes impacting security	
Number of technical notes expiring next month	
Number of projects with issues identified	
Number of projects monitored by IT Security	
Number of internet users by high tiers	Reduce the number of users with elevated access to Internet
Incidents opened and closed in current month	Reduce the cost of resolving IT Security incidents
Percentage of incidents where field intervention was necessary to solve the issue (manual/total)	
Percentage of SAP job functions	Reduce the number of users with SAP Segregation of Duties (SOD) conflicts
Percentage of SAP approved risky profile users	
Number of SAP manual actions	
Incidents opened and closed in current month	Reduce the resolution time for IT Security incidents

focus on.

For example, for the IT service goals *Reduce the cost of resolving IT Security incidents solution,* and *Reduce the time for resolving IT Security incidents,* we analyzed the *Incident Management* process and noticed that manual resolution of incidents was usually more expensive and timedemanding than automatic or remote solutions. Thus, strategies to achieve these goals should focus on avoiding situations that lead to manual resolution of incidents. The analysis of the Incident Management process revealed that manual resolution occurs when the IT Security is unable to automatically remove or remotely resolve a threat. Considering that, we conducted a causal analysis to investigate what causes the inability to automatically remove or remotely resolve a threat. The main identified causes were related to Measurement (Remote support team is not measured by amount of solved incidents), Method (Remote access can only be done with user presence), Machinery (Remote support team does not justify when redirect an incident to local support team) and Manpower (Remote access failure to user's workstation or server). The domain expert informed the among these causes, the one related to Machinery, whose effect avoids remote access necessary for remote solutions, is the most critical. It is related to an error in the default installation of users' workstations that prevents enabling the Remote Procedure Call service. This service is required to

Table 8

SINIS measurement plan for the indicator "Percentage of incidents where field intervention was necessary to solve the issue (manual/total)".

Measurement Plan	
Related IT service Goal	Reduce the cost of resolving IT Security incidents; Reduce the resolution time for IT Security incidents
Measurement Goal	Reduce the rate of manual resolutions for IT Security incidents
Information Need	What is the number of IT Security incidents that needed a manual intervention to be solved?
Measurable Entity	Incidents Management Process
Indicator	Percentage of incidents in which field intervention was necessary to solve the issue (manual/total)
Base Measures	NM = Number of IT Security incidents that needed a manual intervention to be solved
	TN = Total number of IT Security incidents
Measurement Calculation Formula	NM/TN
Measurement Procedure	Extract the incidents list from Remedy application where the Designated Group is IT Security, the Status is solved, and the Summary is antivirus events. Export the incidents list to a spreadsheet in the Microsoft Excel. In the exported spreadsheet, filter and count the number of lines where the Resolution Type was manual (NM), and compute the total number of lines (TN). Store the spreadsheet in the IT Security measurement directory in the current month's folder. Apply the measurement calculation formula and obtain the value of the Percentage of incidents where field intervention was necessary to solve the issue (manual/total).
Measurement Responsible	Incidents Analyst
Measurement Unit	-
Measurement Moment	5th working day of the month
Measurement Periodicity	Monthly

Table 9 SINIS strategy "Enable Remote Procedure Call on Workstations". IT service Goals Reduce the cost of resolving IT Security incidents; Reduce the resolution time for IT Security Incidents Strategy Name Enable Remote Procedure Call on workstations Strategy Scope Create (i) a new installation image to workstations, test it and used it in new workstations, and (ii) a script to enable and start Remote Procedure Call service in every restart in existent workstations. Strategy Owner End User Manager Strategy Sponsor IT Service Director Strategy Complexity Low Strategy Risk Low Strategy Cost No Cost - Usage of internal resources only

Strategy Context FactorsThe End User department would not hire a new service provider to deliver the strategy because there is no available budgetStrategy AssumptionsThe End User department would execute the strategy with existing resources and cannot guarantee when the strategy would be completed. There are
users with personal workstations that do not have standard installation image and do not login in network to run the script.

be enabled and running for a remote access session to be established. Therefore, strategies to achieve the *Reduce the cost of resolving IT Security incidents solution,* and *Reduce the time for resolving IT Security incidents* should consider this issue.

Taking the findings of the processes analysis into account, we performed the **Establish Strategies to Achieve IT service Goals** activity. Considering the previous example, the following strategy was defined: create (i) a new installation image to workstations, test it and use it in new workstations, and (ii) a script to enable and start Remote Procedure Call service on every restart in existing workstations. The defined strategies were documented by using a SINIS template, as shown in Table 9.

Next, the **Identify Indicators for Strategies** activity was performed. During this activity, measures suggested in the COBIT Goals Cascade and in the IT service list of measures were consulted. However, the organization decided to keep using the indicators already in place and define new indicators, not reusing indicators suggested in the COBIT Goals Cascade and in the IT service list of measures. For identifying the indicator related to the strategy presented in Table 9, we asked the domain expert how to verify whether the strategy was successful. He stated that he would need information about the relation between the number of times that remote resolution could not be performed due to Remote Procedure Call not being enabled and the number of manual incidents. Thus, for monitoring the strategy, the indicator described in Table 10 was defined.

5.2.4. Create interpretation models for all indicators

After documenting all the indicators (i.e., indicators related to IT service goals and strategies) the **Create Interpretation Models for all Indicators** phase was initiated to determine how the data collected for the indicators should be interpreted, and to drive decision-making. Table 11 shows as an example the interpretation model established for the indicator described in Table 10.

5.2.5. Build, review and adjust the GQM + Strategies grid

In the last phase, the grid containing all the information produced

during the previous activities was built. The context factors, assumptions, goals, strategies and indicators were organized in a GQM + Strategies Grid and presented to the IT Security coordinator and team to gather feedback. Fig. 4 shows a fragment of the resulting grid.

After applying SINIS, we conducted an interview with the IT Security coordinator aiming to get feedback about SINIS use. Since the IT Security team was overloaded building the single-page report demanded by the board of directors, and the IT Security coordinator acts between the IT Security manager and the IT Security team and communicates with both of them, the IT Security manager designated him to summarize perceptions on behalf of all the study participants. IT Security coordinator is graduated in Computer Science, has 20 years of experience in Information Technology and has played the IT Security coordinator role for 7 years. The research question was used as the starting point for interview and according to the coordinator answers, the supporting questions were used. The interviewe was told to feel free to talk as much they wanted to. The interview took around 60 min and it was recorded and transcribed. Transcript was validated with the coordinator by email.

5.3. Case study analysis, interpretation and lessons learned

In this section, we discuss results about how the research question was answered and we also present some lessons learned that we intend to use in the future to improve SINIS.

5.3.1. Results

When answering the research question (To what extent can SINIS be used to identify goals, strategies and indicators for IT Services?), the coordinator said that by using SINIS, the IT Security team was able to properly define IT service goals aligned to business goals, review existing indicators, keep only indicators aligned to goals, create strategies to achieve the defined goals and indicators to monitor the strategies. Process modeling and causal analysis helped the IT Security team to clarify and understand bottlenecks in the processes and possible root causes for problems that prevent goals achievement. According to the

Table 10

Measurement plan for "Enable Remote Procedure Call on workstations".

Related IT service Goal	Reduce the cost of resolving IT Security incidents; Reduce the time of resolving IT Security incidents
Measurement Goal	Reduce
Information Need	How many times remote resolution was not performed because Remote Procedure Call was not enabled?
Indicator	Percentage of manual incidents caused by remote resolution fail due to not enable Remote Procedure Call
Measurable Entity	Incident Management
Base Measures	NMI = Number of manual incidents caused by remote resolution fail due to not enable Remote Procedure Call
	TN = Total number of manual incidents
Measure Calculation Formula	NMI/TN
Measurement Moment	First day of the month to measure data from previous month
Measurement Periodicity	Monthly
Measurement Procedure	N1: Extract data from the Incidents Report considering the Incident Type and the Incident Resolution field N2: Extract data from the Incidents
	Report considering the Incident Type
Measurement Responsible	Personnel from IT Security responsible for antivirus
Measure Unit	Percentage

Table 11 Interpretation Model.

incipictation model.		
	IT service Goal Indicator	IT service Strategy Indicator
Indicator	Percentage of incidents where field intervention was necessary to resolve the	Percentage of manual incidents where remote support failed due
	issue (manual / total)	to Remote Procedure Call (RPC) not being enabled
IT service Goal/Strategy	IT service Goals: Reduce the cost of resolving IT Security incidents and Reduce the time of resolving IT Security incidents	IT service Strategy: Enable Remote Procedure Call on workstations
Target	20%	10%
Range	Reduction	Reduction
Baseline	60% last year	40% last year
Interpretation	If value is until 5% over target, then only verify isolated cases. If value is more than 5% over target, then review root cause and strategies in place.	If value is until 5% over target, then only verify isolated cases. If value is more than 5% over target, then review implemented strategy.
Interpretation Responsible	IT Security responsible for antivirus	IT Security responsible for antivirus
Interpretation Moment	At the end of each month, starting one month after the End User team completes the strategy	At the end of each month, starting one month after the End User team completes the strategy
Interpretation Periodicity	Every month, the current value is compared to the target and to the previous month as a reference. At the end of the year, the total value is compared to the total value for the previous year.	Every month, the current value is compared to the target and to the previous month, as a reference to verify if the strategy was successful.





coordinator, SINIS also contributed to make clear for the team the relation between goals, strategies and indicators. He stated that after the case study, the IT Security team was more dedicated to measurement activities because members understood the relationship between what they should measure, what they should do (i.e., the strategies), and the relation with the IT service goals. He also reported that some concepts used in SINIS to define goals, strategies, indicators and interpretation models were new to him and the team. He stated that learning about these concepts and documenting goals, strategies and indicators by using SINIS templates helped them better understand what should be done, why it should be done and how it should be done. Moreover, he informed that the team was motivated to keep using SINIS and that the SINIS description and supporting instruments are good specifications that make it possible to execute the activities. The coordinator feedback showed us that, according to him, it was possible to execute SINIS activities based on their descriptions, SINIS produced what it is supposed to deliver, and the use of SINIS provided several benefits for the IT Security department. When asked about the effort demanded to perform SINIS, the coordinator expressed some concern regarding the amount of time required to perform SINIS activities. However, he also said that he believes that the effort tends to decrease as the team becomes more familiar to SINIS. At the end, he said that, taking the obtained results into account, he found the effort to apply SINIS justifiable and acceptable.

We can understand the obtained results as initial evidences that SINIS is feasible (it was possible to execute SINIS by following its description, it produced a set of IT service-related goals, strategies and indicators aligned to organizational goals, and it required efforts considered acceptable) and useful (its use provided benefits for the IT Security department). However, the study presents limitations (discussed in Section 5.4) that prevents results generalization. Thus, although the study provided initial evidences about SINIS feasibility and usefulness, they cannot be generalized.

In addition to answer the research question, after applying SINIS we also evaluated it in terms of its requirements (presented in Section 3).

- Requirement R1 is about identifying goals and indicators at multiple levels. Goals were cascaded from the organization (business goals) to the IT Security department (IT service goals) and derived operational actions (strategies). Indicators were defined to monitor IT Security goals and the strategies defined to achieve them.
- Requirement R2 is about fostering alignment between indicators and goals. According to the IT Security coordinator, defining IT service goals and associating indicators to them was the main benefit of applying SINIS, as they reviewed the existing indicators and discarded indicators not aligned to any goal.
- Requirement R3 is about providing instruments to support SINIS execution. Templates, examples and checklists were used during

B. Trinkenreich et al.

Table 12

Lessons Learned.		
SINIS Characteristic	Lesson Learned	Improvement
Use of different sources to allow reusing elements (e.g., goals and indicators)	(-) SINIS suggests that indicators can be identified by consulting the COBIT Goals Cascade set of indicators or a list of IT service indicators provided in [49]. However, we noticed that searching a catalog can be ineffective for organizations that have a set of indicators being used.	Provide guidelines about how to select or improve indicators used by the organization.
	(-) Search in different sources can be difficult because sources do not follow a common conceptualization and categorization.	Create a unique catalog of IT service goals, strategies and indicators using RMSO conceptualization and categorize these elements according to maturity models' processes, COBIT and ITIL processes, aiming to make easier search and reuse.
	(+) Consulting the indicators from COBIT Goals Cascade and from [49] provided insights for the team refine their own indicators.	-
Use of causal analysis	 (-) Using causal analysis is a good way to find out issues to be addressed by the strategies to be defined to achieve IT service goals. By identifying the aspects that the strategies should focus on, an 	Provide some guidelines about which causal analysis techniques better applies to some situations.
	organization can be prevented from working on many possible (and sometimes inefficient) initiatives. However, SINIS does not support choosing the most suitable causal analysis technique to a given situation. Thus, the decision about which technique to use depends	
SINIS supporting instruments	on the knowledge of who is applying SINIS. (-) SINIS templates, checklists and examples were helpful to apply SINIS. However, more information might be needed for personnel not very familiar with IT service processes or measurement. Also,	Develop a supporting tool and other mechanisms to facilitate SINIS use.
	due to the lack of a supporting tool, some activities demanded much time.(+) SINIS provides templates to be filled during activities execution	-
	and also examples of filled templates. Having available examples to read during each activity helped the team to easily understand what needed to be filled in the provided templates.	
Establishment of numeric targets to indicators	(-) SINIS suggests that numeric targets are established for the defined indicators. Targets were established looking at past experiences and considering the manager expectations. There was no information or analysis about the processes capacity to meet the established targets.	Include in SINIS the use of Statistical Process Control to establish more realistic targets and evaluate if the processes are able to achieve them.
	(+) Establish numeric targets for the indicators was good for having	-
Explicit relations among goals, strategies and indicators	a way to measure if strategies are performing as expected. (+) SINIS helps align goals (what is to be achieved), strategies (how to achieve goals) and indicators (how to monitor goals achievement	-
	and strategies success). By doing that, team members became more aware about why they should work on the established strategies and provide measurement data.	
Measurable Strategies	(+) SINIS guides the use of measurement plans to connect indicators to goals and strategies. When applying the measurement plans for measuring the strategies, the IT service manager was able to gather information to evaluate the strategies and, indirectly, also the performance of the team members performing the strategies. This served as a more transparent and non-subjective performance evaluation that was considered less subjected to complaints and	-

SINIS execution and were considered by the study participants useful to learn how to execute SINIS and to properly record the produced results.

- Requirement R4 is about reusing indicators. Even not reusing indicators from the lists of indicators provided by SINIS, these lists provided insights that helped the IT Security department to reuse and refine some of its own indicators. Moreover, SINIS helped to reduce the time and effort to identify indicators, since in the study the team defined for the first time IT Security goals before thinking about indicators. Thus, indicators identification was strictly focused on measuring goals achievement, instead of involving lengthy unfocused discussions as in the past.
- Requirement R5 is about adopting a consistent terminology. SINIS instruments use terms of the conceptualization provided by a measurement ontology. Once the terms were explained to the participants, they did not have any doubt and we observed no difficulty about the terms meaning during the study.

5.3.2. Lessons learned

The case study allowed us to learn some lessons. In Table 12 we

classify a lesson as positive (+) when it refers to SINIS aspects that supported successful results. On the other hand, we classified a lesson as negative (-) when it refers to aspects that need to be addressed in future improvements of SINIS.

5.4. Threats to validity to the case study results

The validity of a study denotes the trustworthiness of the results, and to what extent the results are not biased by researchers' subjective point of view [44]. In this section, we discuss threats involved in the case study. Every study has threats. They should be addressed as much as possible and be considered together with the results obtained in the study. Next, we present the main threats involved in this study by following the classification proposed in [44], namely: construct, internal, external and reliability.

Construct Validity: this aspect of validity is related to the constructs involved in the study. The main threat in this context is the same discussed in Section 3.1, i.e., indicators we used to evaluate SINIS may not be able to fully represent properties they operationalize. Moreover, there is a threat about constructs discussed in interview questions or

presented in SINIS supporting instruments are not interpreted in the same way by the researcher and study participants. For example, in SINIS' context, the "*strategy*" term has a very particular meaning (like in GQM+Strategies) and the terminology adopted to talk about measurement is very domain-specific. This could lead to misunderstandings. To minimize this threat, the terminology adopted in SINIS and indicators used to evaluate SINIS were explained to study participants and doubts were clarified. Additionally, data collected during the interview were validated by the interviewee.

Internal Validity: this aspect of validity is concerned with the relationship between results and the applied treatment. The main threat in this context is related to the researcher who conducted study. The participation of this researcher can have affected study results, since she has knowledge about SINIS (she is one of its creators) and she works in the organization where the study was conducted. Although she does not work in the department where the study was performed, she has knowledge about the organization's business goals and processes. Thus, we cannot state that same results could be obtained if the researcher had not guided the execution of SINIS activities. Trying to minimize this threat, the researcher limited her interference to guide the participants through the execution of SINIS activities and to suggest the use of SINIS provided checklists, templates and examples to support activities execution. Another threat refers to time constraints faced by the IT Security team during the case study. Since the IT Security team had a tight deadline imposed by the director to present a single-page report with a dashboard including the most relevant indicators, it did not have enough time to fully cascade, during the case study, all the identified IT service goals in strategy. To minimize the effects of time constraints on the study, the researcher asked the participants to select a subset of the identified IT service goals and all SINIS activities were performed to them. By doing that, the study participants experienced all SINIS activities.

External Validity: this aspect of validity is concerned with to what extent it is possible to generalize results. Main threats to external validity in this study are: (i) researcher participation; (ii) use of SINIS in a single department of a single organization; (iii) feedback obtained from only one participant of the study. Concerning (i), as discussed in the context of internal validity, the researcher participation may have influenced results. Thus, it is not possible to generalize results for cases without the researcher intervention. As for (ii), the use of SINIS in a single department provides results that can only be considered for other departments similar to the IT Security department of Organization A. Regarding (iiii), perceptions about SINIS use were taken from only one participant. To minimize this threat, the selected the participant who could summarize perceptions from the IT Security manager and team, since he interacts and communicates with both. Considering these threats, study results can only be generalized to departments similar to the IT Security department of the Organization A that are interested in identifying IT service goals, strategies and indicators aligned to business goals, and where the researcher conducts SINIS use.

Reliability Validity: this aspect is concerned with to what extent data and analysis depend on specific researchers. The researcher participation also represents a threat to reliability validity. The researcher only guided the execution of SINIS activities, i.e., she did not execute the SINIS activities herself, in order to minimize her influence (as a researcher) in SINIS' context of use in produced data, as previously discussed. During SINIS execution, interviews and meetings were performed to gather necessary data to perform SINIS activities. All interviews and meetings were recorded, transcribed, and transcripts were validated by participants and a researcher. Another reliability threat is related to the interview to gather feedback about SINIS use. The interview conducted was not totally structured and the interviewee was let to freely talk about his perceptions about SINIS. To minimize the threat of not getting the necessary information to evaluate SINIS, we defined a set of supporting questions to be used depending on answers provided from the interviewee. The analysis of the interview data

was made by only one researcher, which represents another threat to reliability validity.

In summary, considering all mentioned threats and the fact we have performed a single case study, we can only present some insights regarding SINIS use and generalization is limited. Thus, obtained results cannot be considered conclusive, but preliminary evidences of SINIS feasibility and usefulness.

6. Conclusions, related work and implications

6.1. Conclusions

Being part of the largest economic sector in the world [47], IT services have been growing by adopting an IT management serviceoriented approach to support applications, infrastructure and processes [39]. IT services must be aligned to organizational needs and goals and must be able to contribute to goals achievement. Although the IT service literature states that relevant indicators must be used to verify goals achievement and support decision-making [19], there is a lack of clear guidance about how to define proper indicators aligned to organizational needs and goals [27] [33]. Therefore, IT service departments can spend time and effort measuring without being sure about what the measurement results represent [52].

This paper presented SINIS, a method that extends GQM + Strategies to help organizations identifying goals, strategies and indicators for IT services in alignment with business goals. SINIS consists of a systematic process and instruments (checklists, templates and examples) to support the process execution.

SINIS was used in a case study in the IT Security department of a large global organization. The IT Security team was able to build the GQM+Strategies Grid and discard useless indicators. Team members became more devoted to measurement and strategies, and now understand relationships between goals, strategies and indicators. SINIS instruments contributed to cover the lack of practical examples and guidance to apply GQM+Strategies [6]. The obtained results cannot be considered conclusive, but they show initial evidence that SINIS can be used to identify goals, strategies and indicators for IT services.

SINIS was developed through incremental learning cycles. Each cycle consisted of a study that produced useful knowledge for developing SINIS. In addition to SINIS itself, which is the main contribution of this work, other contributions were produced: (i) a set of measures suitable for IT service measurement [49]; (ii) aspects to be considered when identifying IT service indicators [48]; (ii) knowledge related to analysis of impacts among IT service-related processes [50]; (iii) relevant knowledge about using GQM+Strategies in the IT service domain [51]; and (iv) knowledge involving the use of causal analysis techniques to identify aspects to be addressed when defining strategies to achieve IT service goals [54].

6.2. Related works

As previously discussed, COBIT Goals Cascade and GQM + Strategies are the main basis for SINIS. Thus, there are some similarities between SINIS and these works, but there are also some differences. COBIT Goals Cascade provides a catalog of goals and indicators which can be reused by IT service organizations. COBIT Goals Cascade recommends that each organization should build its own goals cascade, compare it with COBIT Goals Cascade, and then refine it [25]. However, COBIT Goals Cascade does not provide any mechanism to drive this process. As COBIT Goals Cascade, SINIS recommends that goals are cascaded from higher to lower levels. SINIS suggests accessing the COBIT Goal Cascade catalog for reuse and covers the lack of mechanism by providing procedures, checklists, templates and examples that can be used by organizations to define its own goals and indicators. GQM + Strategies recommends that goals, measures and strategies should be aligned and modeled in a grid to make goals and strategies explicit for an

B. Trinkenreich et al.

organization and to provide a clear correlation of all measurement initiatives [6]. However, GQM + Strategies does not detail how to identify processes or issues to be considered in strategies or how to define proper strategies and measures [35]. Similar to GQM + Strategies, SINIS uses a grid to represent goals, strategies and indicators. However, different from GQM + Strategies, SINIS recommends analysis of critical processes and causal analysis a way to define proper strategies and indicators to monitor them.

At the best of our knowledge, besides SINIS, there is no other approach that extends GQM+Strategies to the IT service domain providing a systematic process, checklists, templates and examples. However, there are some research efforts which apply GOM + Strategies and discuss some of the gaps addressed by SINIS. Moreover, there are proposals which, although not devoted to IT services, can be used in this context. For example, López et al. [34], Basili et al. [7] and Kaneko et al. [28] provide lessons learned, results and experiences from applying the GQM+Strategies approach in industry, but they do not suggest any kind of method or procedures to be used when applying GQM+Strategies. Asghari [2] uses action research and proposes an elicitation approach called Goal Strategy Elicitation to support collecting information for goals and strategies to apply GQM + Strategies. That author considers that there is a need to conduct more empirical research on GQM+Strategies, as the approach has been evaluated in few cases thus far.

Concerning research efforts related to identification of indicators for measuring IT services, a framework for measuring IT services is presented by Lepmets et al. [32] and later validated in industry [33], but only a catalog is provided, not a method to define and select indicators aligned to goals. Moreover, the authors state that alignment between business goals and IT services needs to be studied in the industry and could provide additional support for their framework. Jäntti et al. [27] present a support system for IT service measurement. According to the authors, in addition to a well-designed and easy-to-use measuring tool, there is a need for a systematic measurement process, and measures need to be based on business goals. To answer this need, the authors suggest a summarized framework based on ITIL [39], but emphasize that the study focused on implementing the measurement system and that the framework was not validated in real cases.

6.3. Implications

The contributions of this work have implications for both, practice and research. In this section, we discuss some of these implications.

6.3.1. Implications for practice

In the context of practice, SINIS can be used to help IT service organizations or departments to identify IT service goals, strategies and indicators aligned to business goals. By doing that, they will be able to measure only what really matters and obtain useful information for decision making. The use of proper indicators allows monitoring strategies and verifying if they produce the expected results. By monitoring a strategy through effective indicators, it is possible to keep it running, when indicators show that it is producing the expected results; to adjust it, when indicators show that the results are not so good; or abort it, when indicators show that the strategy will not be able to produce what it is expected. Moreover, when strategies are established to achieve IT service goals, it is possible to verify if IT service goals will be achieved by the established strategies. This way, it is possible to review goals and strategies according to information provided by indicators used to monitor them. Additionally, when IT service goals are aligned to business goals, it is possible to analyze how the strategies established to achieve IT service goals contribute to business goals achievement. As a result, the alignment between the IT service area and the organization as a whole is improved.

SINIS is meant to be applied by IT service managers or similar roles. The IT service manager can execute SINIS to identify and cascade goals,

Information and Software Technology xxx (xxxx) xxx-xxx

strategies and indicators and have the big picture about how his team is going to work in a certain period (next year, for example) to achieve business goals. However, it is important to notice that the execution of some SINIS activities can involve other participants. For example, if a causal analysis is needed, may be necessary to involve people directly related to the processes to be analyzed. Moreover, it is important that the results of SINIS application (especially the final grid) are communicated to the interested parties.

Although the expected role to execute SINIS is the IT service manager, SINIS can be also executed by other IT service professionals or even by the project team to have a structured view about which projects, activities, initiatives or practical operational actions (i.e., strategies) can be taken to achieve goals. The structured view of alignment between goals, strategies and indicators can be useful to justify a new project by indicating that it is able to produce results for achieving business goals.

For being able to apply SINIS, we suggest that the user is familiar with IT-related service processes and measurement; however, no knowledge of the approaches used as foundations for SINIS (e.g., COBIT Goals Cascade and Balanced Scorecard) is required. We believe that the process description and the supporting instruments (checklists, templates and examples) allow a person that had never executed SINIS to apply it. However, since, so far, we have not performed any study without the intervention of the researchers, we do not have evidence that the process description and supporting instruments are enough for one to apply SINIS successfully.

The current version of SINIS was applied in an IT service department of a global large organization. The method was designed to be used by IT service departments that could be both a service provider organization (as in the case study) or the organization to which service is provided. In this last case, the organization can apply SINIS to align the provided services to the organization's goals as well as to monitor the provided services. However, we have not explored the use of SINIS in this context yet.

Although we have not experienced it yet, we believe that there is no offside that prevents applying SINIS in a small size organization. Activities can spend more effort for large organizations that probably have more goals, strategies and indicators, or less for small organizations with less goals, strategies and indicators.

SINIS is a proposal strongly related to the industry. However, there are still some steps to be followed aiming at transferring it to the practice. In order to scale SINIS to practice in field, new studies in industry are necessary, particularly without researchers' participation. By conducting new studies, we will be able to evaluate SINIS use in different scenarios, identify and make the necessary improvements to transfer it to the practice. Taking this into account, we plan to perform new case studies to better evaluate SINIS and get feedback for improving it. Moreover, aiming at improvements to foster the use of SINIS in practice, as future works, we plan to develop a tool to support SINIS use and an online catalog to make the list of IT service measures identified in [52] available for automatic search.

6.3.2. Implications for research

Implications of this work are mainly related to the GQM + Strategies [6]. The first one regards the use of that approach in the IT Service domain. GQM + Strategies is an approach that supports identifying goals, defining strategies (i.e., initiatives, such as projects or other actions) to achieve the goals, and identifying measures/indicators to monitor strategies and goals. It was conceived in the Software Engineering area and it is focused on software development. We have explored its use in a different domain by providing a method based on GQM + Strategies structure to be used in the IT Service domain. Another implication related to GQM + Strategies is that SINIS can be considered an extension of that approach as we provide guidelines about how to elicit strategies and instruments (checklists, templates and examples) to support executing each activity. The use of SINIS can bring new issues

B. Trinkenreich et al.

to be explored regarding the use of GQM + Strategies in the IT Service domain. Moreover, the guidelines, checklists, templates and examples represent useful knowledge to apply GQM + Strategies in the IT Service domain and, from their use, new knowledge and needs to be addressed in future researches can arise.

We have explored the extension of the GQM + Strategies to apply it in IT service measurement initiatives, but it is possible to notice that SINIS focuses on the planning phase of a measurement program. Thus, SINIS supports identifying IT service-related goals, strategies and indicators aligned to business goals, but does not address goals and strategies monitoring. In this context, some research questions can be explored in future works, such as: How should goals, strategies and indicators be monitored to verify goals achievement? How should changes in context factors, assumptions and goals should be managed in the context of the defined strategies? New research questions can also be explored when planning IT service-related goals, strategies and indicators: After identifying and aligning goals, strategies and indicators to achieve business goals, how to verify if the identified elements really are the necessary ones for achieving business goals? Are interpretation models enough to monitor, control and report information about goals and strategies? Considering that IT service processes can conflict with one another [50], how conflicts between IT service goals and strategies can be avoided or minimized? Summarizing, our work is just a first step in research to apply GQM+Strategies principles to the IT Service domain. There is still a promising road ahead.

Acknowledgments

The authors wish to thank CAPES, FAPERJ (Projects E-26/010.000883/2016 e E-26/010.001644/2016, E-211.174/2016), CNPq (Process 461777/2014-2) and UNIRIO (Project PQ-UNIRIO 01/2016 and 01/2017) for their financial support.

References

- [1] M. Abdi, P.D. Dominic, Strategic IT alignment with business strategy: service oriented architecture approach, International Symposium on Information Technology. Kuala Lumpur – Malaysia, vol. 3, 2010, pp. 1473–1478, http://dx. doi.org/10.1109/ITSIM.2010.5561624.
- [2] N. Asghari, Evaluating GQM Strategies Framework for Planning Measurement System, Master Degree Thesis School of Computing, Blekinge Institute of Technology (BTH), Karlskrona, Sweden, 2012, p. 54.
- [3] Y. Bartens, F. Schulte, Voss, E-business IT governance revisited: an attempt towards outlining a novel Bi-directional business/IT alignment in COBIT5, 47th Hawaii International Conference on System Sciences (HICSS), 2014, pp. 4356–4365, , http://dx.doi.org/10.1109/HICSS.2014.538.
- [4] Y. Bartens, S. De Haes, Y. Lamoen, F. Schulte, S. Voss, On the way to a minimum baseline in IT governance: using expert views for selective implementation of COBIT 5, 48th Hawaii International Conference on System Sciences (HICSS), 2015, pp. 4554–4563, http://dx.doi.org/10.1109/HICSS.2015.543.
- [5] V.R. Basili, H.D. Rombach, G. Caldiera, Goal question metric paradigm, Encyclopedia of Software Engineering 2 Volume Set John Wiley & Sons, Inc, 1994.
- [6] V. Basili, A. Trendowicz, M. Kowalczyk, J. Heidrich, C. Seaman, J. Münch, D. Rombach, Aligning Organizations Through Measurement – The GQM + Strategies Approach, Springer, 2005, http://dx.doi.org/10.1007/978-3-319-05047-8.
- [7] V. Basili, C. Lampasona, A. Ramírez, Aligning corporate and IT goals and strategies in the oil and gas industry, 14th PROFES – International Conference Product-Focused Software Process Improvement, Paphos, Cyprus, vol. 7983, 2013, pp. 184–198, http://dx.doi.org/10.1007/978-3-642-39259-7_16.
- [8] M.P. Barcellos, R.A. Falbo, A.R. Rocha, A well-founded software process behavior ontology to support business goals monitoring in high maturity software organizations, 2010 14th IEEE International Enterprise Distributed Object Computing Conference Workshops (EDOCW), 2010, pp. 253–262, http://dx.doi.org/10.1109/ EDOCW.2010.15.
- [9] M.P. Barcellos, R.A. Falbo, A.R. Rocha, Establishing a well-founded conceptualization about software measurement in high maturity levels, 2010 7th International Conference on the Quality of Information and Communications Technology (QUATIC), Porto, 2010, pp. 467–472, http://dx.doi.org/10.1109/ QUATIC.2010.84.
- [10] M.P. Barcellos, R.,A. Falbo, R. Dalmoro, A well-founded software measurement ontology, 6th International Conference on Formal Ontology in Information Systems (FOIS 2010), Toronto – Canada. Frontiers in Artificial Intelligence and Applications. Amsterdam: IOS Press, vol. 209, 2010, pp. 213–226, http://dx.doi.org/10.3233/ 978-1-60750-535-8-213.
- [11] M.P. Barcellos, R.A. Falbo, A.R. Rocha, Using a reference domain ontology for

Information and Software Technology xxx (xxxx) xxx-xxx

developing a software measurement strategy for high maturity organizations, 2012 16th IEEE International Enterprise Distributed Object Computing Conference Workshops (EDOCW), Beijing, 2012, p. 114, , http://dx.doi.org/10.1109/EDOCW. 2012.24.

- [12] S. Barr, Practical Performance Measurement Using the PuMP Blueprint for Fast, Easy and Engaging Performance Measures, in: N. Wishner (Ed.), first ed., 2014.
- [13] M. Cases, D.A. Bondner, B. Mutnury, Architecture of Service Organizations, in: G. Salvendy, W. Karwowski (Eds.), Introduction to Service Engineering, John Wiley & Sons, Inc., Hoboken, 2010, pp. 109–134.
- [14] P.F. Drucker, The Practice of Management: A Study of the Most Important Function in America society, Harper Collins, New York, 1954, p. 368.
- [15] M. Dumas, M. La Rosa, J. Mendling, H.A. Reijers, Qualitative Process Analysis in Fundamentals of Business Process Management, Springer, Berlin Heidelberg, 2013, pp. 185–211, http://dx.doi.org/10.1007/978-3-642-33143-5_6.
- [16] W. Eckerson, Performance Dashboards: Measuring, Monitoring and Managing Your Business, second ed., Wiley, 2011 ISBN 047091842X, 9780470918425.
- [17] K. Engelmann, Service science-where practice meets theory, Service Science: Fundamentals, Challenges and Future Developments, Springer, Berlin, 2008, pp. 119–136, http://dx.doi.org/10.1007/978-3-540-74489-4_12 part 4.
- [18] W.A. Florac, R.E. Park, A.D. Carleton, Practical software measurement: Measuring for process management and improvement (No. CMU/SEI-97-HB-003), Carnegie-Mellon Univ Pittsburgh pa Software Engineering Institute, 1997.
- [19] E. Forrester, B. Buteau, S. Shrum, CMMI For Services, Guidelines for Superior Service. CMMI-SVC Version 1.3, second ed., SEI. Addison-Wesley Professional, 2010 ISBN 0132700484, 9780132700481.
- [20] F. García, M.F. Bertoa, C. Calero, A. Vallecillo, F. Ruiz, M. Piattini, M. Genero, Towards a Consistent Terminology for Software Measurement Information and Software Technology, Inf. Software Technol. 48 (8) (2006) 631–644.
- [21] J.G. Guzmán, H.A. Mitre, A. Amescua, M. Velasco, Integration of strategic management, process improvement and quantitative measurement for managing the competitiveness of software engineering organizations, Software Qual. J. 18 (2010) 341–359.
- [22] A.R. Hevner, A three cycle view of design science research, Scand. J. Inf. Syst. 19 (2007) 87–92 http://aisel.aisnet.org/sjis/vol19/iss2/4 Last accessed on March 2018.
- [23] V. Hrgovcic, W. Utz, D. Karagiannis, Service modeling: a model based approach for business and IT alignment, 35th IEEE Annual Computer Software and Applications Conference Workshops. Munich, Germany, 2011, http://dx.doi.org/10.1109/ COMPSACW.2011.77.
- [24] C.M. Huxley, An Improved Method to Identify Critical Processes, Faculty of Information Technology. Masters by Research thesis Queensland University of Technology, Brisbane, 2003.
- [25] ISACA, COBIT 5 control objectives management guidelines maturity models: enabling processes, Information Systems Audit and Control Association, USA, 2012.
- [26] ISO/IEC, ISO/IEC 20.000-1: information technology service management part 1: service management system requirements, International Standard Organization / International Electrotechnical Commission, Switzerland, 2011.
- [27] M. Jäntti, A. Lahtela, J. Kaukola, Establishing a measurement system for IT service management processes: a case study, Int. J. Adv. Syst. Measure. 3 (3 & 4,) (2010), http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.671.8462&rep= rep1&type=pdf, Last accessed on March 2018.
- [28] T. Kaneko, M. Katahira, Y. Miyamoto, M. Kowalczyk, Application of GQM + Strategies in the Japanese space industry, Joint Conference of the 21st International Workshop on and 6th International Conference on Software Process and Product Measurement (IWSM-MENSURA), 2011, pp. 221–226, http://dx.doi. org/10.1109/IWSM-MENSURA.2011.22.
- [29] R. Kaplan, D.P. Norton, The Balanced Scorecard Translating Strategy Into Action, Harvard Business School Press, Boston, 1996 ISBN 0875846513, 9780875846514.
- [30] T. Kilpi, Implementing a software measures program at Nokia, IEEE Software 18 (6) (2001) 72–77, http://dx.doi.org/10.1109/IWSM-MENSURA.2011.22.
- [31] M. Kowalczyk, H. Barthel, J. Münch, J. Heidrich, A. Trendowicz, A deployment process for strategic measurement systems, 8th Software Measurement European Forum (SMEF), Rome, Italy, 2011, pp. 45–60 https://arxiv.org/abs/1312.1040 Last accessed on March 2018.
- [32] M. Lepmets, E. Ras, A. Renault, A quality measurement framework for IT services, Annual SRII Global Conference, San Jose, CA, 2011, pp. 767–774, http://dx.doi. org/10.1109/SRII.2011.84.
- [33] M. Lepmets, A.L. Mesquida, A. Cater-Steel, A. Mas, E. Ras, The evaluation of the IT service quality measurement framework in industry, Global J. Flexible Syst. Manage. 15 (1) (2014) 39–57, http://dx.doi.org/10.1007/s40171-013-0052-7.
- [34] G. López, B. Aymerich, D. Garbanzo, A. Pacheco, Application of GQM + Strategies in a multi-industry state-owned company, PROFES – International Conference on Product-Focused Software Process Improvement, 2016, pp. 198–214, http://dx. doi.org/10.1007/978-3-319-49094-6_13.
- [35] G. López, A. Pacheco, F. Cocozza, D. Garbanzo, B. Aymerich, GQM + Strategies and IDEAL: a combination of approaches to achieve continuous SPI, PROFES – International Conference on Product-Focused Software Process Improvement, 2016, pp. 311–326, http://dx.doi.org/10.1007/978-3-319-49094-6_20.
- [36] P.P. Maglio, S.L. Vargo, N. Caswell, J. Spohrer, The service system is the basic abstraction of service science, Inf. Syst. e-Business Manage. 7 (4) (2009) 395–406, http://dx.doi.org/10.1007/S10257-008-0105-1.
- [37] J. Mcgarry, D. Card, C. Jones, B. Layman, E. Clark, J. Dean, F. Hall, Practical Software Measurement: Objective Information For Decision Makers, first ed., Addison Wesley, Boston, USA, 2002 ISBN 0-201-71516-3.
- [38] J. Munch, F. Fagerholm, P. Kettunen, M. Pagels, J. Partanen, Experiences and insights from applying GQM+Strategies in a systems product development

organisation, 39th Euromicro Conference Series on Software Engineering and Advanced Applications, 2013, pp. 70–77, , http://dx.doi.org/10.1109/SEAA. 2013.62.

- [39] OGCa, ITIL service operations, The Stationary Office TSO. London, UK, 2011.
- [40] K. Petersen, C. Gencel, N. Asghari, S. Betz, An elicitation instrument for operationalising GQM+Strategies (GQM+S-EI), Emp. Software Eng. 20 (4) (2015) 968–1005, http://dx.doi.org/10.1007/s10664-014-9306-z.
- [41] A. Parasuraman, L.A. Zeithaml, L.L. Berry, A conceptual model of service quality and its implications for future research, J. Marketing 49 (1985) 41–50, http://dx. doi.org/10.2307/1251430.
- [42] D. Parmenter, Key Performance Indicators Developing, Implementing and Using Winning KPIs, third ed., Wiley, 2015, p. 448 ISBN 1118925106, 9781118925102.
- [43] D. Robitaille, Root Cause Analysis: Basic Tools and Techniques, first ed., Paton Press, 2004, p. 104 ISBN 1932828028, 9781932828023.
- [44] P. Runeson, M. Host, A. Rainer, B. Regnell, Case Study Researcy.h in Software Engineering: Guidelines and Examples, John Wiley & Sons, Hoboken, NJ., 2012, p. 256 ISBN 111818100X, 9781118181003.
- [45] R. Steinberg, Measuring ITSM Measuring, Reporting, and Modeling the IT Service Management Measures That Matter Most to IT Senior Executives, Trafford Publishing, 2013, p. 196 ISBN 149071944X, 9781490719443.
- [46] D. Steuperaert, Improving the quality of the COBIT 5 goals cascade as an IT process prioritisation mechanism, Int. J. IT/Business Alignment Governance (IJITBAG) 7 (2) (2016) 49–69, http://dx.doi.org/10.4018/IJITBAG.2016070104.
- [47] J.M. Tien, D. Berg, On services research and education, J. Syst. Sci. Syst. Eng. 15 (3) (2006) 257–283, http://dx.doi.org/10.1007/s11518-006-5019-1.
- [48] B. Trinkenreich, G. Santos, Evaluation of measurement process for incidents, continuity and availability management under the light of MR-MPS-SV maturity model, Portuguese: Avaliação do Processo de Medição para Serviços de TI em uma Empresa Global à Luz do MR-MPS-SV), 10th Annual MPS Program Workshop (WAMPS), Campinas, Brazil, 2014.

[49] B. Trinkenreich, G. Santos, M. Barcellos, Metrics to support IT service maturity

Information and Software Technology xxx (xxxx) xxx-xxx

- models a systematic mapping study. J7th International Conference on Enterprise Information Systems (ICEIS), Barcelona, Spain, vol. 2, 2015, pp. 330–337, , http:// dx.doi.org/10.5220/0005376003300337.
- [50] B. Trinkenreich, G. Santos, Metrics to support IT service maturity models a case study, 17th International Conference on Enterprise Information Systems (ICEIS), Barcelona, Spain, vol. 2, 2015, pp. 395–403, http://dx.doi.org/10.5220/ 0005398003950403.
- [51] B. Trinkenreich, G. Santos, Evaluation of incident management process under the light of MR-MPS-SV maturity model and measurement to support IT service quality improvement, Portuguese: Avaliação da Gerência de Incidentes sob a Luz do MR-MPS-SV e Medição para Apoiar a Melhoria da Qualidade do Serviço de TI), 14th Brazilian Symposium on Software Quality (SBQS), Manaus, Brazil, 2015.
- [52] B. Trinkenreich, G. Santos, M. Barcellos, SINIS a method to select indicators for IT services, 16th International Conference on Product-Focused Software Process Improvement (PROFES), Bolzano, Italy, vol. 9459, 2015, pp. 68–86, , http://dx.doi. org/10.1007/978-3-319-26844-6_6.
- [53] B. Trinkenreich, G. Santos, M.P. Barcellos, A Method to Select Goals, Indicators and Strategies for IT Services, (2017) Technical Report DIA/UNIRIO 003/2007. Available at http://www.seer.unirio.br/index.php/monografiasppgi/article/view/ 6509.
- [54] B. Trinkenreich, G. Santos, M.P. Barcellos, T. Conte, Eliciting strategies for the GQM + Strategies approach in IT service measurement initiatives, Empirical Software Engineering and Measurement (ESEM) International Symposium, Toronto, Canada, 2017, http://dx.doi.org/10.1109/ESEM.2017.51.
- [55] R. Wieringa, Design Science Methodology For Information Systems And Software Engineering, Springer, Heidelberg, 2014, p. 332 ISBN 978-3-662-43839-8.
- [56] R. Wieringa, M. Daneva, Six strategies for generalizing software engineering theories, Sci. Comput. Program 101 (2015) 136–152, http://dx.doi.org/10.1016/j. scico.2014.11.013.